Author's response to reviews

Title: Implementing evidence-based medicine in general practice: a focus group based study

Authors:

Karin Hannes (Karin.hannes@med.kuleuven.be)
Marcus Leys (mleys@gesg.vub.ac.be)
Etienne Vermeire (Etienne.Vermeire@uiw.ua.ac.be)
Bert Aertgeerts (Bert.aertgeerts@med.kuleuven.be)
Frank Buntinx (Frank.buntinx@med.kuleuven.be)
Anne-Marie Depoorter (AMDepoorter@vub.ac.be)

Version: 4 Date: 15 August 2005

Author's response to reviews: see over
Covering letter for the article with ID 693703765014609,

Dear editors,

My co-authors and I thank you for accepting our article. Please find below the final revisions made to the latest version.

1. We adjusted the list of authors in the submission system according to the manuscript and removed the titles in the submission system.
2. We deleted the phrase on equal contributions.
3. We incorporated the footnote in the text and added an explanation on the incompatibility of numbers in table 6.
4. We added the references from the tables in the main text, using square brackets.
5. We replaced the underlined subheadings with bold ones and doubled checked on the Actor-Factor titles (which we think are correct)
6. We changed all tables into regular grids without colour and avoided tabbed text. We highlighted with bold.
7. We included a new item ‘figures’ in the main text and uploaded the figure as a separate file.

We hope that this final version of the article will be appreciated by the editorial team. We apologize for any delay in resubmitting this version due to a stay in China.

Yours sincerely,

On behalf of the team of authors,

Karin Hannes
Belgian Centre for Evidence-Based Medicine
E-mail: Karin.hannes@med.kuleuven.be